Author's response to reviews

Title: Patients visiting the complementary medicine clinic for pain: a cross sectional study

Authors:

Roni Peleg (pelegr@bgu.ac.il)
Orly Liberman (orlylib@bgu.ac.il)
Yan Press (yanp@inter.net.il)
Pesach Shvartzman (spesah@bgu.ac.il)

Version: 4 Date: 27 April 2011

Author's response to reviews:

April 27, 2011

Editor
BMC Complementary and Alternative Medicine

MS -1514463964912508: "Patients visiting the complementary medicine clinic for pain: a cross sectional study"

Dear Editor:

In response to the additional revisions requested attached is the corrected version of our manuscript.

1. Full details of the ethics committee that granted ethical approval for the study was included in the methods section.
2. Competing interests declaration was added in the manuscript before the authors' contributions section.
3. Highlighting was removed from all manuscript text.

Please do not hesitate to contact me for any further questions or clarifications.

I look forward to your reply.

Sincerely,

Roni Peleg, MD.
Corresponding Author